Author's response to reviews

Title: Clinical response to Auron Misheil therapy treatment in a patient with advanced multifocal hepatocellular carcinoma: a case report

Authors:

Jürgen S Scheele (juergen.scheele@uniklinik-freiburg.de)
Jan Harder (jan.harder@hbh-kliniken.de)
Zoran Stankovic (zoran.stankovic@uniklinik-freiburg.de)
Daniel Räpple (daniel.raepple@uniklinik-freiburg.de)
Annette Dorn (anette.dorn@uniklinik-freiburg.de)
Hans C Spangenberg (hans.spangenberg@uniklinik-freiburg.de)
Hubert E Blum (hubert.blum@uniklinik-freiburg.de)

Version: 5 Date: 16 August 2011

Author's response to reviews: see over
Dear JMCR Editorial Team,

Please find enclosed our revised manuscript. We inserted now the sentence "Consent was obtained from the patient's wife and not from the patient himself because he died in the meanwhile" in the Consent section of the manuscript (highlighted in blue).

Please let me know if there is any further information required.

I am looking forward to hearing from you.

Yours sincerely,

Prof. Dr. Dr. Jürgen Scheele